

JGR Biogeosciences

RESEARCH ARTICLE

10.1029/2023JG007927

Key Points:

- Dust deposition, indicated by high trace element concentrations, is higher closest to the delta of the Ä'äy Chù' into Lhù'ààn Mân'
- Evidence for dust deposition at Lhù'ààn Mân' can be found in trace elements of the 0–2 cm soil layer and lichens influenced by distance
- Vegetation characteristics are impacted by dust deposition more than soil characteristics

Supporting Information:

Supporting Information may be found in the online version of this article.

Correspondence to:

S. Pouillé, sophie.pouille@umontreal.ca

Citation:

Pouillé, S., Talbot, J., Tamalavage, A. E., Kessler-Nadeau, M. É., & King, J. (2024). Impacts of mineral dust on trace element concentrations (As, Cd, Cu, Ni and Pb) in lichens and soils at Lhù'ààn Mân' (Yukon Territory, Canada). *Journal of Geophysical Research: Biogeosciences*, *129*, e2023JG007927. https://doi.org/10. 1029/2023JG007927

Received 22 NOV 2023 Accepted 16 MAY 2024

Author Contributions:

Conceptualization: Sophie Pouillé, Julie Talbot, James King Data curation: Sophie Pouillé Formal analysis: Sophie Pouillé, Anne E. Tamalavage Funding acquisition: Julie Talbot, James King Investigation: Sophie Pouillé, Julie Talbot, Max Émile Kessler-Nadeau, James King Project administration: Sophie Pouillé, Julie Talbot, James King

© 2024. The Author(s). This is an open access article under the terms of the Creative Commons Attribution-NonCommercial-NoDerivs License, which permits use and distribution in any medium, provided the original work is properly cited, the use is non-commercial and no modifications or adaptations are made.

Impacts of Mineral Dust on Trace Element Concentrations (As, Cd, Cu, Ni and Pb) in Lichens and Soils at Lhù'ààn Mân' (Yukon Territory, Canada)

Sophie Pouillé¹, Julie Talbot¹, Anne E. Tamalavage², Max Émile Kessler-Nadeau¹, and James King¹

¹Département de Géographie, Université de Montréal, Montréal, QC, Canada, ²Département des Sciences Biologiques, Université de Montréal, Montréal, QC, Canada

Abstract Dust is a mineral aerosol of the atmosphere that often contains trace elements such as As, Cd, and Pb. Lhù'ààn Mân' (Kluane Lake), located in southwestern Yukon, is a region of frequent dust activity. In 2016, the lake level fell due to a dramatic decrease in inflow from glacier meltwater, and the delta of the lake became an important source of dust to surrounding ecosystems. To determine the impacts of dust deposition on vegetation and soil trace element concentrations and characteristics, we sampled the lichen *Peltigera canina* and soil layers at 57 sites along a deposition gradient located 1.4–33.6 km downwind from the principal dust source. Arsenic, Cd, Cu, Ni and Pb in lichens were negatively correlated with the distance away from the dust source, with the highest correlations in Ni and Pb ($r^2 = 0.50$ and 0.48, respectively). Lichen and tree abundances were negatively impacted by dust deposition, suggesting that dust can affect ecosystem vegetation composition. Starting 8 km away from the dust source, the concentrations of As, Ni, and Pb decreased by more than 50% per km, while Cd and Cu concentrations decreased by more than 40% per km. Overall, within the sampled ecosystems, soil pH is 1.4 times higher in the first 8 km from the dust source while carbon content and nutrients are lower, which implies changes in nutrient availability and cycling in dust-affected ecosystems.

Plain Language Summary The delta of Lhù'ààn Mân' (Kluane Lake), located in southwest Yukon and on the traditional lands of Kluane First Nation, Champagne-Aishihik First Nation, and White River First Nation, is an area where dust often settles and climate change has accelerated the melting of glaciers since the last ice age. Due to reduced glacier water flowing into the lake, the sediments in the delta are more exposed to wind, creating dust storms. To understand the impacts of dust deposition within the surrounding ecosystems, we assessed the levels of trace metals in lichens and soils across a gradient from the delta to the lake. We found that distance from the dust source affected trace metal concentrations in lichens, surface soils, and vegetation. It is essential to understand how these metals spread as they can be potentially toxic and harmful to ecosystems and humans.

1. Introduction

Atmospheric dust originates from human activities (mining, gravel roads, etc.) and natural processes (geological and soil erosion, glacier melt) (Bullard et al., 2016; Carling et al., 2017; Csavina et al., 2012; Ravi et al., 2011). Active glaciers produce fine sediments transported by fluvial and aeolian processes as dust (Anderson et al., 2017; Bullard et al., 2016). High latitude cold environments (defined as north of 50°N and south of 40°S) contribute at least 5% of the global dust budget, although their contribution to global dust circulation is seasonally variable (Bullard, 2013; Bullard et al., 2016; Mahowald et al., 2006; Meinander et al., 2022). Glaciers are retreating in most parts of the world due to climate change, and, thus, the valleys they currently occupy could become an important dust source in the coming decades (Bullard et al., 2016).

Dust deposition affects several ecological and biogeochemical processes on scales that range from individual plants to the entire globe (Field et al., 2010). Dust deposition also influences biogeochemical cycles of nutrients (carbon C, nitrogen N, phosphorus P) and trace elements (TE) (e.g., arsenic, copper, lead) in terrestrial and aquatic ecosystems (Bachelder et al., 2020; Das et al., 2013; Formenti et al., 2011; Muhs, 2013; Neff et al., 2008; Swap et al., 1992). For instance, dust deposition represents an important source of N and P in nutrient-limited ecosystems such as the forests in the Amazon and Hawaii (Chadwick et al., 1999; Okin et al., 2004). Dust deposition also appears to impact fundamental terrestrial ecosystem functions such as vegetation composition, soil

the Term



Resources: Julie Talbot, Anne E. Tamalavage, Max Émile Kessler-Nadeau, James King Supervision: Julie Talbot, Anne E. Tamalavage, James King Validation: Sophie Pouillé, Julie Talbot, Anne E. Tamalavage, James King Visualization: Sophie Pouillé Writing – original draft: Sophie Pouillé, Julie Talbot, Anne E. Tamalavage, James King Writing – review & editing: Anne E. Tamalavage, Max Émile Kessler-Nadeau properties, and nutrient cycles (Muhs et al., 2007; Myers-Smith et al., 2006; Okin et al., 2006; Walker & Everett, 1987).

Lichens have been used for monitoring spatial and/or temporal patterns of atmospheric deposition (Bargagli, 2016; Carignan et al., 2002; Klapstein et al., 2020; Simonetti et al., 2003; Sloof, 1995) and for studying longrange transport of trace elements in high latitudes (e.g., Riget et al., 2000; Singh et al., 2013; Wojtuń et al., 2013). Lichens are widespread in boreal and arctic regions, and are slow-growing, perennial organisms that maintain a uniform morphology through time (Bergamaschi et al., 2004; Seaward, 2008). Lichens can accumulate TE and, depending on the lichen species and specific element, can be sensitive to TE exposure (Bargagli, 2016; Nash, 2008; Neitlich et al., 2022; Tyler, 1989). Lichens can also record past and present TE accumulation. For example, Bargagli et al. (2002) showed high arsenic concentrations in lichens (*Parmelia caperata*) near mining and smelting operations in a former industrial district in Italy. Additionally, *Xanthoria parietina* lichens in Southwest France accumulated arsenic and lead from coal combustion during the early twentieth century (Agnan et al., 2013).

The Dhal T'à' (AÄ'äy Chù' Valley) that connects to Lhù'ààn Mân' (Southern Tutchone name of Kluane Lake) is a high latitude active proglacial landscape with high dust emissions during the spring and summer in the Ä'äy Chù' River delta, and has been impacted by climate change with the rapid retreat of Kaskawulsh glacier since 2016 (Bachelder et al., 2020; Nickling, 1978; Shugar et al., 2017). This dust is rich in TE including arsenic (As), cadmium (Cd), copper (Cu), nickel (Ni) and lead (Pb) (Bachelder et al., 2020). These TE may come from minerals, as they are present in rocks from the Kluane ultramafic belt or from processes (e.g., rupturing of clay producing fine particulate matter) that releases TE when eroded by wind (Bachelder et al., 2020; Hulbert, 1997; Masuda, 2018).

The goal of this work was to determine the impacts of TE-rich dust deposition on TE concentrations in lichens and soils in a remote area, and to elucidate the implications of this dust deposition for vegetation and soil properties of dust-receiving ecosystems. To do so, we used TE concentration within the microlichen *Peltigera canina* as our main indicator of dust deposition, collected at sites following a depositional gradient around Lhù'ààn Mân'. This lichen of the Peltigeraceae family is associated with cyanobacteria (*Nostoc* spp.), widespread across boreal and subarctic regions (Martínez et al., 2003; Miadlikowska & Lutzoni, 2000) and has a high growth rate (Alam et al., 2015; Beckett et al., 2013; Paulsrud et al., 2001). Lichens have been mostly used to monitor anthropogenic TE deposition in both urban and remote areas (Agnan et al., 2013; Bargagli, 2016; Carignan et al., 2002; Klapstein et al., 2020; Naeth & Wilkinson, 2008; Parviainen et al., 2019; Widory et al., 2015; Zdanowicz et al., 2017). In an active dust region like Lhù'ààn Mân', it is likely that dust emission would result in higher TE concentrations in *Peltigera* lichens and in underlying soils at closer proximity to the dust source.

2. Materials and Methods

2.1. Site Description

Lhù'ààn Mân' ($61^{\circ}13'03''$ N, $138^{\circ}37'34''$ W, Figure 1a) is located in the Yukon territory between the Ruby Ranges to the northeast and the Kluane Ranges in the St. Elias mountains on the west (Figures 1a and 1b). These lands are traditional territories of Kluane, Champagne-Aishihik, and White River First Nations. The climate is semi-arid, and the area receives a mean annual precipitation of 275 mm, mostly during the summer, with a mean annual temperature of $-2.5^{\circ}C \pm 1.1$ (1991–2020 climate normal, Canada, E. and C. C., 2011; LaZerte & Albers, 2018), as measured at Burwash Landing. Typically, ~105 cm of snow falls annually, but it is variable dependent on altitude. Wind velocity and air turbulence can increase transport of sediment in regions where winds often go above 5 m/s (Bullard et al., 2016; Nickling, 1978). At Lhù'ààn Mân', there are strong off-glacial winds (2.0–6.0 m/s) mainly coming from the southeast (Figure 1c). These winds erode the fine sediments of the delta and transport them across the southern part of the lake (Bachelder et al., 2020; Nickling, 1978).

The region is part of the boreal forest biome with white spruce (*Picea glauca*) as the dominant tree type (Crofts et al., 2018; Krebs et al., 2001). The understory is dominated by soapberry (*Shepherdia Canadensis*), and a well-developed ground layer composed of herbs (e.g., *Lupinus arcticus, Linnaea borealis, Hedysarum alpinum, Geocaulon lividum*), lichens (e.g., *Peltigera canina, Collema tenax, Cladonia* spp.) and forest floor mosses such as *Pleurozium schreberi* and *Ptilium crista-castrensis* (Crofts et al., 2018; Douglas & Vitt, 1976; Fremlin

10.1029/2023JG007927

21698961,

2024, 6, Downloaded from https://agupubs



Figure 1. Maps of sampling sites in the Lhù'ààn Mân' (Kluane Lake) region. The three figures are oriented north. (a) Location of the study site Lhù'ààn Mân' in Canada, the boreal forest is defined as the broad, circumpolar vegetation zone of high northern latitudes mainly covered with forests and other cold-tolerant wood species (Brandt, 2009); (b) Location of the sampling sites in Dhal T'à' (Ä'äy Chù' Valley) and around Lhù'ààn Mân'; (c) Prevailing winds recorded at the weather station of Kluane Lake Research Station (KLRS) between 2017 and 2023 (https://webapp.ccadi.ca/kluane).

et al., 2011; Harris, 1987; Hoefs, 1976; Hoefs & Thomson, 1972; Marsh et al., 2006; Vetter, 2000). Soils within this region have high pH, high calcium carbonate, and an A-horizon rich in organic matter (Laxton et al., 1996). The high calcium carbonate content at the surface reflects the incorporation of neoglacial calcareous loess from the unvegetated floodplains of the Dhal T'à' since deglaciation, about 12,500 B.P (Brahney et al., 2008; Denton & Stuiver, 1967; Hoefs, 1976; Laxton et al., 1996; Marsh et al., 2006). There is also the presence of discontinuous and scattered permafrost soils in the area (Krebs et al., 2001). Sampling sites (orange dots, Figure 1b) in the Dhal T'à' (Ä'äy Chù' Valley) and on the west part of the lake are on top of sandstone, limestone, conglomerate and argillite (fine grained sedimentary rocks with high clay content) (Israel & Cobbett, 2008; Muller, 1967). These geologic deposits are rich in silica, aluminum, iron, calcium and magnesium minerals. Fine-grained sedimentary rocks such as sandstone, argillite and conglomerate can contain high arsenic concentrations (Masuda, 2018; Tanaka, 1988). High arsenic concentrations are often associated with deposits such as copper, cadmium, lead and nickel (Tanaka, 1988). The glacial till is an eroded feature of those deposits and is the primary source of dust. Sampling sites around the lake (yellow dots, Figure 1b) are within glacial till deposits.

2.2. Sampling

To measure TE concentrations associated with dust deposition and assess its spatial extent, we sampled the terricolous lichen *Peltigera canina* L. Willd. in August 2021 with a stainless-steel knife and scissors, in three replicas, at 57 sites around the lake to capture dust deposition. Sampling sites were located near the roads for accessibility reasons, but the sampling was done at least 100 m from the road, as road dust can affect arctic and subarctic ecosystems especially within the first 50 m from the road (Li et al., 2023; Myers-Smith et al., 2006). In addition to the lichen sampling, two soil layers were sampled, one at 0–2 cm and the other at 2–10 cm in three replicas at each site.

We characterized the percent cover of each vegetation layer (tree, shrub, herb, moss, and lichen layers) within a 20-m radius around the central point of each sampling site. We followed the inventory method recommended by the *Field Manual for Describing Terrestrial Ecosystems* (Erwin, 2010), which is used by the territory (Environment Yukon, 2016, 2017). The tree layer includes at least the three dominant woody species (Lachance et al., 2021). The shrub layer includes at least 90% of woody plants less than 4 m tall; the herb layer includes all herbaceous species; and the lichens and moss layer include all bryophytes and terrestrial lichens occurring on rock and wood (Erwin, 2010).

2.3. Lichen and Soil Analyses

To prepare for the lichen analyses, 200 mg of dried and ground lichen was digested on a block digester in 2 mL of concentrated certified trace metal grade nitric acid (HNO₃) at room temperature for 12 hr and then at 80°C for at least 12 hr. For the analysis of the soils, 25 mg of dried-milled soil was digested in a block digester in 4 mL of concentrated certified trace metal grade HNO₃ and 1 mL of concentrated certified trace metal grade hydrochloric acid (HCl) at room temperature for 12 hr and then at 80°C for at least 12 hr. Triplicate samples were pooled before analysis. Supernatants were filtered at 0.45 µm. Trace elements (As, Cd, Cu, Ni, and Pb) were determined in lichens and soils and minor elements (Al, Ca, Fe, K, Mg, and Mn) in soils by Inductively coupled plasma mass spectrometry (ICP-MS; PerkinElmer NexION 300×, PerkinElmer Syngistix). The ICP-MS analysis run was randomized. We used procedural and instrumental blanks to quantify a potential contamination during the digestions and analyses. The Ontario Geological Survey 1878P certified fen peat reference material (Riley, 1989), QCS-3 and SCP-QC4, were used in quality assurance and control procedures for the lichens with a recovery rate between 94% and 100.8% (Table S1 in Supporting Information S1). SCP-QC3 and QCS-27 was used for the soils with a recovery rate between 92.5% and 114.0% (Tables S2 and S3 in Supporting Information S1).

Total C and N content in lichens and soils were analyzed on 2 and 5 mg of dried-milled samples, respectively, by CHNO-S elemental analysis (Fisons - EA-1108 CHNS-O Element Analyzer). Total P content was analyzed using a Shimadzu UV spectrophotometry using a colorimetry assay based on the Murphy and Riley method (Murphy & Riley, 1962). We prepared a soil solution with a ratio of 1:10 for the 0–2 cm layer and a ratio of 1:2 for the 2–10 cm layer to measure the pH (Fisher Accumet pH Meter AB 150).

2.4. Statistical and Spatial Analyses

Statistical analyses were performed using *R* version 4.2.2 (R Core Team, 2022) and RStudio 2023.6.1.524 (Posit team, 2023). All data was assessed for normality using Shapiro-Wilk tests because the data should be as close as possible to the condition of normality for parametric analyses (Borcard et al., 2018). The relationships between variables influencing TE concentrations in lichens and soils were evaluated using parametric Pearson's (r^2) correlations to compare non dimensionally homogeneous quantitative variables (Borcard et al., 2018). To determine the changes in ecosystem functions linked to dust deposition, redundancy analyses (RDA) were performed using site characteristics (slope, orientation of the site, and distance) and TE concentrations in lichens (As, Cd, Cu, Ni, and Pb) with environmental variables following methods within Fritze et al. (1989).

Ordinary exponential kriging was considered a suitable way to represent TE concentrations at Lhù'ààn Mân' because it is the principal geostatistical technique that can be used to predict TE concentration in un-sampled locations (Orton et al., 2009; Wackernagel, 2003). Kriging is a statistically based estimator of spatial variables with three components (Bolstad & Manson, 2022). The first component is the spatial trend that depends on the direction, the second component describes the local spatial autocorrelation and the third component is random, stochastic variation (Bolstad & Manson, 2022). All ordinary exponential kriging was performed in RStudio, using the version of *R* previously described. The settings are available in Table S4 in Supporting Information S1. All the maps were made using QGIS 3.30.1 (QGIS Development Team, 2023).

3. Results

3.1. Trace Elements in Lichens

Trace element concentrations in lichens decrease with distance from the dust source (Figures 2 and 3). Indeed, the correlation analysis shows that the distance from the dust source as the most important factor influencing TE concentrations in lichens (Figure S1a in Supporting Information S1). At 8 km or more from the dust source, TE concentrations decrease by more than 50% per km, except for Cu and Cd (Table S5 in Supporting Information S1). Among the TE, As, Ni, and Pb have the strongest statistically significant relationships with distance from the dust source ($r^2 = 0.44$, 0.50, and 0.48, respectively; p < 0.01), while concentrations of Cd and Cu have relatively weaker but still significant relationships with distance from the dust source ($r^2 = 0.38$ and 0.25 respectively; p < 0.001; Figure 3).



Journal of Geophysical Research: Biogeosciences

10.1029/2023JG007927



Figure 2. Prediction maps of trace elements concentrations in lichens (*Peltigera canina*) based on ordinary interpolation. Concentration levels are classified into five intervals. Note that the interval range of values differs by trace element.

3.2. Trace Elements in Soils

Arsenic, Ni and Pb concentrations in the 0-2 cm layer decrease with distance from the dust source. This suggests distance has an influence on TE concentrations in the 0-2 cm layer (Figure 4). In contrast, TE concentrations in the 2-10 cm layer are not correlated with distance except for Cu concentrations (Figure 5).

3.3. Impacts on Soil Characteristics

Within this work, we categorize soil characteristics as follows: pH, base cations (Ca, K, Mg, Na), iron (Fe), aluminum (Al), carbon (C), nitrogen (N) and phosphorus (P). The RDA of the 0–2 cm layer characteristics indicates that 37.7% of the variation in the 0–2 cm layer characteristics is explained by TE concentrations in lichens, distance from the dust source, elevation, slope, and orientation of the site (p = 0.001; Figure 6a). Axis 1 explains 62.8% of the variance (p = 0.001) and is most closely associated with Cd concentrations and distance. Axis 2





Figure 3. Trace elements concentrations in lichens (mg/kg) versus distance from the dust source (km). R² and p values are derived from the non-spatial linear regressions. Dashed lines depict statisticallysignificant curves. The shading shows the 95% confidence interval.

explains 21.4% of the variance (p = 0.011) and is associated with orientation and slope. Concentrations of As, Cu, Ni, and Pb in lichens and elevation contribute more to axis 1 as indicated by higher eigenvalues ($\lambda = 5.39$ and $\lambda = 1.83$ respectively).



Figure 4. Plots of trace element concentrations in 0–2 cm layer (mg/kg) versus distance from the dust source (km). R and p values are derived from the non-spatial linear regressions. Dashed lines depict statistically significant curves. The shading shows the 95% confidence interval.





Figure 5. Plots of trace element concentrations in the 2–10 cm layer (mg/kg) versus distance from the dust source (km). R and p values are derived from the non-spatial linear regressions. Dashed lines depict statistically significant curves. The shading shows the 95% confidence interval.

However, the RDA of the 2–10 cm layer characteristics shows that 27.9% of the overall variation in 2–10 cm characteristics is explained by the environmental variables (p = 0.001; Figure 6b). Axis 1 explains 51% of the variance (p = 0.001) and is associated with distance from the source, Pb and Ni concentrations in lichens. Axis 2 explains 24.9% of the variance (p = 0.017) and is associated with slope and orientation. Concentrations of As, Cu, and Cd in lichens, and elevation contribute more to axis 1 ($\lambda = 3.62$) than to the axis 2 ($\lambda = 1.76$).

Soil characteristics are also influenced by the distance from the dust source. pH of the soil shows negative correlations with distance in both the 0–2 cm and 2–10 cm layers (Figures 6a and 6b). However, this negative correlation is weaker in the 2–10 cm layer compared to the 0–2 cm layer, suggesting pH of the 0–2 cm layer is more influenced by dust deposition (Figures 6a and 6b). In the 0–2 cm layer, carbon concentrations increase with distance and decrease with TE in lichens whereas in the 2–10 cm layer, carbon concentrations decrease with distance from the dust source (Figures 6a and 6b). As for nutrients, N and P increase with distance from the dust source in both the 0–2 cm layers (Figures 6a and 6b).

3.4. Impacts on Vegetation Characteristics

The RDA of the vegetation shows that 11.08% of the overall variance in vegetation layers is explained by environmental variables, namely TE concentrations in lichens, distance from the dust source, elevation, slope, and orientation of the site (p = 0.01; Figure 7a). Axis 1 explains 63.7% of the variance (p = 0.01) and is most closely associated with Cd concentrations and distance. Axis 2 explains 22.4% of the variance and is associated with slope and orientation, although these last two variables are not significant (p = 0.7). Additionally, concentrations of As, Ni and Pb, along with elevation contribute more to axis 1, as indicated by a higher eigenvalue ($\lambda = 0.80$) compared to the eigenvalue of axis 2 ($\lambda = 0.25$). As a result, lichen and tree cover exhibit a negative correlation to TE concentrations while shrub and herb cover are positively correlated with TE concentrations.

21698961





Figure 6. Triplots of redundancy analysis (RDA, scale 2) showing the influence of environmental variables (site characteristics and trace elements in lichens, red lines) on (a) 0–2 cm and (b) 2–10 cm layer characteristics (response variables, orange squares) at 57 sites (bullets).

The RDA between the C and macronutrient chemistry of lichens and environmental variables shows that 26.9% of the overall C, N, and P variation is explained by the environmental variables (p = 0.001; Figure 7b). Among these variables, concentrations of As, Cu, Ni and Pb contribute to 55.37% of this variance (p = 0.003) while slope and

8 of 17



Journal of Geophysical Research: Biogeosciences

10.1029/2023JG007927



Figure 7. Relationships between vegetation characteristics and environmental variables. (a) Triplot of the redundancy analysis (scale 2) showing the influence of environmental variables (site characteristics and TE in lichens, red lines) on vegetation cover (response variables, orange squares) at 57 sites (bullets). (b) Triplot of redundancy analysis (scale 2) showing the influence of environmental variables (site characteristics and TE in lichens, red arrows) on carbon, nitrogen, and phosphorus concentrations (response variables, grey squares). (c) Plots of carbon (%) in lichens versus TE concentrations in lichens (mg/kg). Dashed lines represent statistically significant relationships. The shading corresponds to confidence interval (0.95).

orientation contribute to 42.3% (p = 0.034). Cadmium concentrations in lichens, distance from the dust source, and elevation contribute more to axis 1, indicated by a higher eigenvalue ($\lambda = 1.07$) than the eigenvalue of axis 2 ($\lambda = 0.82$). Trace element concentrations negatively influence C and N concentrations in lichens whereas P concentrations, CN and NP ratios are weakly or not correlated. Distance from the dust source is positively correlated to C and N concentrations. Specifically, carbon concentrations in lichens decrease as TE concentrations in lichens increase (Figure 7c). This correlation is stronger for As, Ni, and Pb ($r^2 = 0.70$, 0.73 and 0.74 respectively) than the correlation between Cd and C ($r^2 = 0.56$) and Cu and C ($r^2 = 0.58$).



4. Discussion

4.1. Trace Elements in Lichens and Soils

With the exception of Cu, TE concentrations are all below 5.0 mg/kg in our vegetation samples (Figure S2 in Supporting Information S1), which indicates that our sites are exposed to limited atmospheric pollution compared to other studies on fugitive dust in mining (Neitlich et al., 2017, 2022; Simonetti et al., 2003). Therefore, we interpret TE signatures to be regionally produced in the Lhù'ààn Mân' region, rather than transported from distal sources. We have found increased concentrations of TE within *Peltigera* lichens closer to the dust point locale (the exposed Lhù'ààn Mân' delta). Based on results from kriging interpolation (Figure 2), As, Cd, Cu, Ni, and Pb concentrations in lichens decrease with distance from the exposed delta. This suggests that there is less dust deposition with increased distance from the source, which can be due to the physical distance from the dust source, or other factors like wind characteristics and size of dust particles (Huang et al., 2021; Myers-Smith et al., 2006; Walker & Everett, 1987). Dust emission mainly depends on the wind velocity giving sufficient kinetic energy to remove fine-grained and coarse material from soil and transport them (Brahney et al., 2014). In Lhù'ààn Mân', wind erosion impacts the rupture of clay mineral coatings and transports them a short distance (Bachelder et al., 2020; Nickling, 1978).

TE concentrations in lower soil layers were not correlated with TE concentrations in lichens and/or distance from source (Figure 3), with the exception of Cu concentration, which was positively correlated with distance. The enrichment of Cu in the layer (2–10 cm) with distance could be due to a leaching from high Cu concentrations in the 0–2 cm layer. Indeed, the 0–2 cms of the Lhù'ààn Mân' region are mostly composed of silt and carbonates with calcium, a competitive ion, which might not allow a high retention of Cu (He et al., 2006; Hoefs, 1976; Laxton et al., 1996; Steinnes & Friedland, 2006). Cu is also more abundant in the soil and dust samples (Figure S2 in Supporting Information S1) because of the presence of native copper deposits in the region (Muller, 1967). Thus, localized biogeochemical factors control the observed patterns of Cu concentrations at Lhù'ààn Mân'.

4.2. Impacts on Additional Soil Characteristics

At Lhù'ààn Mân', the 0–2 cm soil characteristics are not only influenced by TE deposition (Axis 1, Figure 6a) but also by landscape features such as slope, elevation, and orientation (Axis 2, Figure 6b). Dust deposition can have an important effect on pH in 0–2 cm horizons through alkalinization and leachate (Gill et al., 2014; Myers-Smith et al., 2006; Walker & Everett, 1987). We found an increased pH closer to the dust source (Figure 6a) which suggests an alkalinization of the 0–2 cm layer induced by dust deposition. However, this finding is not supported by higher amounts of total Ca as well as other cations close to the dust source, except for Mg (Figure 6a). This could be due to other factors controlling soil pH like the presence of geologic deposits low in Ca at sites close to the dust source (orange dots; Figure 1b).

Dust deposition can also reduce the accumulation of organic matter (OM) in soils as a result of changes in vegetation composition, ecosystem productivity, concentrations of exchangeable base cation, and pH (Auerbach et al., 1997; Johnston & Johnston, 2004; Whittinghill & Hobbie, 2012). In the 0–2 cm layer, carbon concentrations are higher at sites located further away from the dust source (Figure 6a), potentially due to a dilution effect or a direct effect of dust deposition on OM accumulation. Forbes (1995) showed short term responses of tundra vegetation abundance, contributing to organic matter content, and soils to aeolian dust deposition. Dust is likely to accumulate into larger pools in the organic soil horizons over time (Neitlich et al., 2017) and can lead to decrease in the organic matter content over several years. The highest C concentrations are located at sites where elevation is the highest (Figure 6a) likely indicating a lower OM decomposition at sites at high elevation (Badía et al., 2016; Griffiths et al., 2009). Thus, dust deposition is not the only driver of OM content in Lhù'ààn Mân' soils.

In the 2–10 cm layer, the dominant influences on soil characteristics are elevation and slope (Figure 6b) indicating that the 2–10 cm layer characteristics do not depend on dust deposition but mainly depend on landscape features like underlying geology. For instance, the negative relationship between pH in these soil layers and slope (Figure 6b) suggests an accumulation of bases that might have been leached from the top (Ibanga et al., 2008; Khan et al., 2013). We observed an unexpected decrease of base cation (Ca, K and Na) concentrations in the 0–2 cm and 2–10 cm soil layers when closer to the dust source (Figures 6a and 6b) whereas other studies found an increase of base cations with dust deposition (Johnston & Johnston, 2004; Myers-Smith et al., 2006). This decrease could be due to a higher water content accelerating the leaching of Ca, K and Na in the soil layers closer



to the dust source (Che et al., 2021; Harris, 1990; Sauer et al., 2007). In the Ä'äy Chù' delta, the soluble salts are dominated by Ca, with K and Na in lesser amount (Harris, 1990; Nickling, 1978). These salts tend to be concentrated at the surface in early summer as the soils dry out and can bond to soil particles (Harris, 1990; Nickling, 1978). The observed decrease in Ca, K and Na concentrations in the soils could also be explained by runoff of meltwater in the spring or wind transport of salts along with the silt loam particles during erosion of loess (Harris, 1990). High concentrations of Ca, K and Mg would suggest that surface soil characteristics are strongly influenced by dust deposition, while the characteristics of the sub-surface horizons are more likely to depend on the weathering of the local bedrock and landscape features (Johnston, 2001; Johnston & Johnston, 2004).

4.3. Impacts on Vegetation Characteristics

Dust deposition can affect vegetation composition and productivity of arctic and mountainous ecosystems (Auerbach et al., 1997; Gill et al., 2014; Müllerová et al., 2011). Yet, it is known that elevation can also exert a strong control on vegetation distribution (Normand et al., 2009). Tree coverage usually decreases with elevation in response to low temperatures and decrease in tree growth (Griffiths et al., 2009; Testolin et al., 2020). Low tree abundances with higher concentrations of As, Cd, Cu, Ni and Pb would suggest that the vegetation patterns in Lhù'ààn Mân' are also influenced by dust deposition (Figure 7a).

Lichens and evergreen tree species are highly sensitive to dust and TE deposition (Conti & Cecchetti, 2001; Cornelissen et al., 2001; Myers-Smith et al., 2006). In Lhù'ààn Mân', TE concentrations in *Peltigera* lichens are good indicators of their exposure to dust deposition. The negative relationship between trace element concentrations with lichen and tree abundances suggests higher TE concentrations might induce lichen and tree losses within the ecosystem (Figure 7a). However, dust may also impact vegetation growth through shading and mechanical effects, limiting photosynthesis activity (Cornelissen et al., 2001; Mandre & Ots, 1999).

Previous studies assessed the impact of calcareous dust deposition from gravel roads on acidophilic vegetation in the arctic tundra, finding losses of about 50% of lichens (*Cladina* spp and *Peltigera* spp) and evergreen species (Gill et al., 2014; Hope et al., 1991; Moorhead et al., 1996; Myers-Smith et al., 2006; Walker & Everett, 1987) with increased TE from dust. These losses were mainly due to pH and shade increase for lichens, and physical damage or bark chemistry changes for evergreen species (Farmer, 1993; Gill et al., 2014; Myers-Smith et al., 2006; Santelmann & Gorham, 1988; Walker & Everett, 1987). Direct smothering can also cause physical damage to all plant species when exposed to dust (Neitlich et al., 2022).

Shrubs and graminoids are more resilient to dust deposition and variations in soil chemistry (Farmer, 1993; Myers-Smith et al., 2006; Walker, 1996). The negative relationship observed between shrub and herb abundances with distance in Lhù'ààn Mân' (Figure 7a) provides evidence that dust is likely increasing shrub and graminoid growth at sites closer to the dust source. The colonization of graminoids and shrub species might alter N and P availability and allow for higher turnover rates within the local boreal forest ecosystem in which biogeochemical conditions are already limited by nutrient availability (Farmer, 1993; Forbes et al., 2001; Myers-Smith et al., 2006; Walker & Everett, 1987). Additionally, closed canopies can negatively impact shrub and herb abundances in forests through excess shade or varied vegetation successions (Bret-Harte et al., 2001; Chapin et al., 1995; Gill et al., 2014; McTainsh & Strong, 2007; Myers-Smith et al., 2006; Walker & Everett, 1987).

Many moss species are especially sensitive to dust and TE deposition (Auerbach et al., 1997; Farmer, 1993; Hope et al., 1991; Myers-Smith et al., 2006). For instance, peatland mosses (*Sphagnum* sp) have been shown to be very sensitive to calcareous and metal deposition from roads in the Alaskan Tundra and in New Brunswick, and to changes in soil pH (Auerbach et al., 1997; Farmer, 1993; Myers-Smith et al., 2006; Walker & Everett, 1987). In this context, mosses can be less abundant or even die after calcareous atmospheric deposition (Hope et al., 1991; Myers-Smith et al., 2006; Walker & Everett, 1987). However, we found no correlation between moss abundance and distance (Figure 7a and Figure S1a in Supporting Information S1). This suggests that the boreal forest mosses found at our sites (*Pleurozium schreberi* and *Ptilium crista-castrensis*) are less sensitive to dust and TE deposition than expected, but they can be sensitive to other environmental variables (Tobias & Niinemets, 2010). This lower sensitivity of bryophytes than lichens to dust and TE deposition was also reported at Red Dog in northern Alaska (Neitlich et al., 2022). We also found a negative relationship between mosses and shrubs, and mosses and herbs (Figure 7a and Figure S1a in Supporting Information S1) indicating that competition between species, access to the light, and litter fall could have a bigger impact on mosses growth in the boreal forest of Lhù'ààn Mân' than dust deposition (Tobias & Niinemets, 2010).

21698961,



Dust can be an important contributor of rock-derived nutrients (base cations and P) to ecosystems in arid and semi-arid regions (Arvin et al., 2017; Brahney et al., 2014). In our study, concentrations of P in lichens were not correlated to distance (Figure 7b) which could indicate that dust does not bring P into vegetation at Lhù'ààn Mân' despite the ability of *Peltigera* lichens to accumulate high P concentrations in their thalli (Asplund & Wardle, 2017) or that the trends shown are insignificant relative to the input expected from dust. For example, P concentrations are lower (average of 443 mg kg⁻¹) in Lhù'ààn Mân' soils compared to P concentrations (3,000 mg kg⁻¹) in dust and lake sediments (1,000–5,000 mg kg⁻¹) in the Wind River Range watershed in Wyoming reported by Brahney et al. (2014). Additionally, we found an increase in P concentrations with distance in 0–2 cm and 2–10 cm layers indicating that dust does not bring P into soils Lhù'ààn Mân'. This increase is also correlated with a decrease in pH in 0–2 cm and 2–10 cm layers (Figure 6) suggesting that this patterns may result from acidification, which can enhance chemical weathering of minerals and contribute to P availability (Marsh et al., 2006).

Lichens derive carbon from algal and cyanobacterial photobionts, and are dependent on the availability of nutrients like N, deposited on the thallus or fixed by cyanobacterial lichens (Dahlman et al., 2003). We found carbon and N concentrations in lichens increase with distance from the dust source (Figure 7b) suggesting that N is not deposited in the ecosystem. The high N concentrations measured in lichens in our study might be explained by a large amount of N fixed by *Peltigera* lichens which often occurs in arctic environments (Asplund & Wardle, 2017; Gunther, 1989) thus increasing C concentrations and C assimilation capacity (Palmqvist et al., 2002).

Despite lichens ability to accumulate trace elements, their accumulation can sometimes have a negative effect on photosynthetic activity (Brown & Beckett, 1984). We found that C concentrations in lichens decrease with increased TE concentrations (Figure 7c) suggesting that TE have a negative effect on lichens likely by reducing thallus size, inducing colour darkening colour and causing cell damages (Goyal & Seaward, 1981; Nagy et al., 2005; Pisani et al., 2011). Arsenic concentrations found at Lhù'ààn Mân' are in the range of concentrations from 0.1 to 10 mg/kg that are sufficient to cause physiological stress to lichens (Pisani et al., 2011). Furthermore, Cu concentrations between 20 and 100 mg/kg are considered as excessive contents for uptake and can cause damage to lichens, bryophytes, and vascular plants (Kabata-Pendias, 2010). Toxicity of a trace element depends on the concentration and its forms. Trivalent As (AsIII) is 5–10 times more toxic than pentavalent As (AsV) (Gamberg, 2000). While we did not look at the speciation of arsenic, there are still toxicity concerns with the level of As found within the soils and lichens if exposure continues. Arsenic can accumulate in the liver, kidney, muscle, skin and hair and cause toxic effects for wildlife and humans (Gamberg, 2000; Ratnaike, 2003). Lead may be absorbed by plants through the leaves or from the soil through the roots (Gamberg, 2000; Kabata-Pendias, 2010).

Lastly, we also found that N decreased with increased TE concentrations (Figure 7b) which suggests that TE accumulation might have a negative influence on N_2 fixation by cyanobacteria, induced by reductions in density of cyanobacteria in lichens (Nagy et al., 2005; Pisani et al., 2011). Thus, the chemical composition of dust might impact lichen growth more than the physical deposition of particles.

5. Conclusions

Dust storms are more intense and frequent in the Lhù'ààn Mân' region since 2016 (Bachelder et al., 2020; Huck et al., 2023; Shugar et al., 2017). Dust depositions were evaluated by measuring TE concentrations in lichens downwind from the dust source in the region. Trace element concentrations in lichens and distance from the dust source, were negatively correlated suggesting that trace elements in lichens were higher near the dust source, specifically between 1 and 4 km. Beyond 8 km from the Lhù'ààn Mân' delta, these concentrations decrease by over 50% per km. The corroboration of TE in lichens and distance allowed us to determine that using relative TE concentrations in lichens was a suitable approach for evaluating the impact of dust in the region. Arsenic, Cd, and Pb performed better to evaluate dust deposition in this region when compared to Ni and Cu, which are found in the local bedrock. Trace element concentrations in 2–10 cm layer. TE concentrations in 2–10 cm horizons were likely more influenced by the chemical composition of the antecedent bedrock or other geochemical processes like leaching. Overall, soil pH, C content, and nutrients in 0–2 cm and 2–10 cm horizons are the main characteristics impacted by dust deposition in the region.



Vegetation was the main component of the ecosystem affected by dust deposition. Lichens and trees, which are more sensitive to dust, were negatively impacted, while shrubs and herbs were positively impacted. Dust deposition has been shown in this study to impact ecosystem characteristics, specifically on vegetation composition and soil chemistry. Thus, this study contributes to a better understanding of the impacts of dust on trace element concentrations, vegetation, and soil characteristics in glacial regions affected by glacial retreat and climate change.

Data Availability Statement

All the data used to determine the impacts of dust on subarctic terrestrial ecosystems in the study are available in the Pangaea repository (Pouillé, 2024): https://doi.pangaea.de/10.1594/PANGAEA.965325.

References

- Agnan, Y., Séjalon-Delmas, N., & Probst, A. (2013). Comparing early twentieth century and present-day atmospheric pollution in SW France: A story of lichens. *Environmental Pollution*, 172, 139–148. https://doi.org/10.1016/j.envpol.2012.09.008
- Alam, M. A., Gauslaa, Y., & Solhaug, K. A. (2015). Soluble carbohydrates and relative growth rates in chloro-cyano- and cephalolichens: Effects of temperature and nocturnal hydration. New Phytologist, 208(3), 750–762. https://doi.org/10.1111/nph.13484
- Anderson, N. J., Saros, J. E., Bullard, J. E., Cahoon, S. M. P., McGowan, S., Bagshaw, E. A., et al. (2017). The arctic in the twenty-first century: Changing biogeochemical linkages across a paraglacial landscape of Greenland. *BioScience*, 67(2), 118–133. https://doi.org/10.1093/biosci/ biw158
- Arvin, L. J., Riebe, C. S., Aciego, S. M., & Blakowski, M. A. (2017). Global patterns of dust and bedrock nutrient supply to montane ecosystems. *Science Advances*, 3(12), eaao1588. https://doi.org/10.1126/sciady.aao1588
- Asplund, J., & Wardle, D. A. (2017). How lichens impact on terrestrial community and ecosystem properties. *Biological Reviews*, 92(3), 1720–1738. https://doi.org/10.1111/brv.12305
- Auerbach, N. A., Walker, M. D., & Walker, D. A. (1997). Effects of roadside disturbance on substrate and vegetation properties in arctic tundra. *Ecological Applications*, 7(1), 218–235. https://doi.org/10.1890/1051-0761(1997)007[0218:EORDOS]2.0.CO;2
- Bachelder, J., Cadieux, M., Liu-Kang, C., Lambert, P., Filoche, A., Galhardi, J. A., et al. (2020). Chemical and microphysical properties of windblown dust near an actively retreating glacier in Yukon, Canada. Aerosol Science and Technology, 54(1), 2–20. https://doi.org/10.1080/ 02786826.2019.1676394
- Badía, D., Ruiz, A., Girona, A., Martí, C., Casanova, J., Ibarra, P., & Zufiaurre, R. (2016). The influence of elevation on soil properties and forest litter in the Siliceous Moncayo Massif, SW Europe. *Journal of Mountain Science*, 13(12), 2155–2169. https://doi.org/10.1007/s11629-015-3773-6
- Bargagli, R. (2016). Moss and lichen biomonitoring of atmospheric mercury: A review. Science of the Total Environment, 572, 216–231. https:// doi.org/10.1016/j.scitotenv.2016.07.202
- Bargagli, R., Monaci, F., Borghini, F., Bravi, F., & Agnorelli, C. (2002). Mosses and lichens as biomonitors of trace metals. A comparison study on Hypnum cupressiforme and Parmelia caperata in a former mining district in Italy. *Environmental Pollution*, 116(2), 279–287. https://doi.org/ 10.1016/S0269-7491(01)00125-7
- Beckett, R. P., Zavarzina, A. G., & Liers, C. (2013). Oxidoreductases and cellulases in lichens: Possible roles in lichen biology and soil organic matter turnover. *Fungal Biology*, 117(6), 431–438. https://doi.org/10.1016/j.funbio.2013.04.007
- Bergamaschi, L., Rizzio, E., Giaveri, G., Profumo, A., Loppi, S., & Gallorini, M. (2004). Determination of baseline element composition of lichens using samples from high elevations. *Chemosphere*, 55(7), 933–939. https://doi.org/10.1016/j.chemosphere.2003.12.010
- Bolstad, P., & Manson, S. (2022). GIS fundamentals: A first text on geographic information systems (7th ed.). Elder Press.
- Borcard, D., Gillet, F., & Legendre, P. (2018). Numerical ecology with R. Springer.
- Brahney, J., Ballantyne, A. P., Kociolek, P., Spaulding, S., Otu, M., Porwoll, T., & Neff, J. C. (2014). Dust mediated transfer of phosphorus to alpine lake ecosystems of the Wind River Range, Wyoming, USA. *Biogeochemistry*, 120(1), 259–278. https://doi.org/10.1007/s10533-014-9994-x
- Brahney, J., Clague, J. J., Menounos, B., & Thomas, W. D. E. (2008). Timing and cause of water level fluctuations in Kluane Lake, Yukon Territory, over the past 5000 years. *Quaternary Research*, 70(2), 213–227. https://doi.org/10.1016/j.yqres.2008.05.001
- Brandt, J. P. (2009). The extent of the North American boreal zone. Environmental Reviews, 17(NA), 101–161. https://doi.org/10.1139/A09-004
 Bret-Harte, M. S., Shaver, G. R., Zoerner, J. P., Johnstone, J. F., Wagner, J. L., Chavez, A. S., et al. (2001). Developmental plasticity allows betula nana to dominate tundra subjected to an altered environment. Ecology, 82(1), 18–32. https://doi.org/10.1890/0012-9658(2001)082[0018: DPABNT]2.0,CO:2
- Brown, D. H., & Beckett, R. P. (1984). Uptake and effect of cations on lichen metabolism. *The Lichenologist*, 16(2), 173–188. https://doi.org/10. 1017/S0024282984000323
- Bullard, J. E. (2013). Contemporary glacigenic inputs to the dust cycle. *Earth Surface Processes and Landforms*, 38(1), 71–89. https://doi.org/10. 1002/esp.3315
- Bullard, J. E., Baddock, M., Bradwell, T., Crusius, J., Darlington, E., Gaiero, D., et al. (2016). High-latitude dust in the Earth system. Reviews of Geophysics, 54(2), 447–485. https://doi.org/10.1002/2016RG000518
- Canada, E. and C. C., (2011). Historical data—climate—environment and climate change Canada. Retrieved from https://climate.weather.gc.ca/ historical_data/search_historic_data_e.html
- Carignan, J., Simonetti, A., & Gariépy, C. (2002). Dispersal of atmospheric lead in northeastern North America as recorded by epiphytic lichens. *Atmospheric Environment*, *36*(23), 3759–3766. https://doi.org/10.1016/S1352-2310(02)00294-7
- Carling, G. T., Rupper, S. B., Fernandez, D. P., Tingey, D. G., & Harrison, C. B. (2017). Effect of Atmospheric Deposition and Weathering on Trace Element Concentrations in Glacial Meltwater at Grand Teton National Park, Wyoming, U.S.A. Arctic Antarctic and Alpine Research, 49(3), 427–440. https://doi.org/10.1657/AAAR0016.071
- Chadwick, O. A., Derry, L. A., Vitousek, P. M., Huebert, B. J., & Hedin, L. O. (1999). Changing sources of nutrients during four million years of ecosystem development. *Nature*, 397(6719), 491–497. https://doi.org/10.1038/17276

Acknowledgments

We acknowledge that the sample collection in this work was done on the Traditional Territory of the Kluane First Nation and the Champagne and Aishihik First Nations, as well as the White River First Nation territories, with their permissions obtained through a sampling permit (21-35S&E) as required by the Yukon. We also acknowledge Parks Canada for providing permissions for sampling in the Kluane National Park and Reserve (permit KLU-2021-38964). This work would not have been possible without the help of Laurent Bourassa-Frégeau in the laboratory. This work was supported by the Université de Montréal, the Fonds de Recherche du Québec-Nature et Technologies (B2X-217534), the Natural Science and Engineering Research Council of Canada (BRD and RGPIN-2016-05417), and the Canadian Mountain Network, a Canadian Government Network of Centers of Excellence (PV143493-NCE)



- Chapin, F. S., Shaver, G. R., Giblin, A. E., Nadelhoffer, K. J., & Laundre, J. A. (1995). Responses of Arctic Tundra to Experimental and Observed Changes in Climate. *Ecology*, 76(3), 694–711. https://doi.org/10.2307/1939337
- Che, M., Gong, Y., Xu, M., Kang, C., Lv, C., He, S., & Zheng, J. (2021). Effects of elevation and slope aspect on the distribution of the soil organic carbon associated with Al and Fe mineral phases in alpine shrub–meadow soil. *Science of the Total Environment*, 753, 141933. https://doi.org/ 10.1016/j.scitotenv.2020.141933
- Conti, M. E., & Cecchetti, G. (2001). Biological monitoring: Lichens as bioindicators of air pollution assessment a review. *Environmental Pollution*, 114(3), 471–492. https://doi.org/10.1016/S0269-7491(00)00224-4
- Cornelissen, J. H. C., Callaghan, T. V., Alatalo, J. M., Michelsen, A., Graglia, E., Hartley, A. E., et al. (2001). Global change and arctic ecosystems: Is lichen decline a function of increases in vascular plant biomass? *Journal of Ecology*, 89(6), 984–994. https://doi.org/10.1111/j. 1365-2745.2001.00625.x
- Crofts, A. L., Drury, D. O., & McLaren, J. R. (2018). Changes in the understory plant community and ecosystem properties along a shrub density gradient. Arctic Science, 4(4), 485–498. https://doi.org/10.1139/as-2017-0026
- Csavina, J., Field, J., Taylor, M. P., Gao, S., Landázuri, A., Betterton, E. A., & Sáez, A. E. (2012). A review on the importance of metals and metalloids in atmospheric dust and aerosol from mining operations. *Science of the Total Environment*, 433, 58–73. https://doi.org/10.1016/j. scitotenv.2012.06.013
- Dahlman, L., Persson, J., Näsholm, T., & Palmqvist, K. (2003). Carbon and nitrogen distribution in the green algal lichens Hypogymnia physodes and Platismatia glauca in relation to nutrient supply. *Planta*, 217(1), 41–48. https://doi.org/10.1007/s00425-003-0977-8
- Darnajoux, R., Lutzoni, F., Miadlikowska, J., & Bellenger, J.-P. (2015). Determination of elemental baseline using peltigeralean lichens from Northeastern Canada (Québec): Initial data collection for long term monitoring of the impact of global climate change on boreal and subarctic area in Canada. Science of the Total Environment, 533, 1–7. https://doi.org/10.1016/j.scitotenv.2015.06.030
- Das, R., Evan, A., & Lawrence, D. (2013). Contributions of long-distance dust transport to atmospheric P inputs in the Yucatan Peninsula. Global Biogeochemical Cycles, 27(1), 167–175. https://doi.org/10.1029/2012GB004420
- Denton, G. H., & Stuiver, M. (1967). Late Pleistocene Glacial Stratigraphy and Chronology, Northeastern St Elias Mountains, Yukon Territory, Canada. GSA Bulletin, 78(4), 485–510. https://doi.org/10.1130/0016-7606(1967)78[485:LPGSAC]2.0.CO;2
- Douglas, G. W., & Vitt, D. H. (1976). Moss-Lichen Flora of St. Elias-Kluane Ranges, Southwestern Yukon. The Bryologist, 79(4), 437–456. https://doi.org/10.2307/3241938
- Environment Yukon. (2017). Southern Lakes Boreal Low Subzone (BOLsl): A field guide to ecosite identification. Retrieved from https://yukon. ca/en/southern-lakes-boreal-low-subzone-bolsl-field-guide-ecosite-identification
- Erwin, C. (2010). Terrestrial Ecosystems Information Guidelines for Developing a Project Plan. *Ministry of Environment and Climate Change Strategy*. https://www.env.gov.bc.ca/fia/documents/TEI_prj_plan_guidelines_apr2010.pdf
- Farmer, A. M. (1993). The effects of dust on vegetation—A review. *Environmental Pollution*, 79(1), 63–75. https://doi.org/10.1016/0269-7491 (93)90179-R
- Field, J. P., Belnap, J., Breshears, D. D., Neff, J. C., Okin, G. S., Whicker, J. J., et al. (2010). The ecology of dust. Frontiers in Ecology and the Environment, 8(8), 423–430. JSTOR. https://doi.org/10.1890/090050
- Flynn, N., & Francis, S., & Environment Yukon. (2016). Yukon ecological and landscape classification and mapping guidelines. Retrieved from https://yukon.ca/sites/yukon.ca/files/env/env-yukon-ecological-landscape-classification-mapping-guidelines.pdf
- Forbes, B. C. (1995). Tundra Disturbance Studies, III: Short-term Effects of Aeolian Sand and Dust, Yamal Region, Northwest Siberia. Environmental Conservation, 22(4), 335–344. JSTOR. https://doi.org/10.1017/s0376892900034901
- Forbes, B. C., Ebersole, J. J., & Strandberg, B. (2001). Anthropogenic Disturbance and Patch Dynamics in Circumpolar Arctic Ecosystems. *Conservation Biology*, 15(4), 954–969. JSTOR. https://doi.org/10.1046/j.1523-1739.2001.015004954.x
- Formenti, P., Schütz, L., Balkanski, Y., Desboeufs, K., Ebert, M., Kandler, K., et al. (2011). Recent progress in understanding physical and chemical properties of African and Asian mineral dust. Atmospheric Chemistry and Physics, 11(16), 8231–8256. https://doi.org/10.5194/acp-11-8231-2011
- Fremlin, K. M., McLaren, J. R., DeSandoli, L., & Turkington, R. (2011). The Effects of Fertilization and Herbivory on the Phenology of the Understory Vegetation of the Boreal Forest in Northwestern Canada. Arctic Antarctic and Alpine Research, 43(3), 389–396. https://doi.org/10. 1657/1938-4246-43.3.389
- Fritze, H., Niini, S., Mikkola, K., & Mäkinen, A. (1989). Soil microbial effects of a Cu-Ni smelter in southwestern Finland. Biology and Fertility of Soils, 8(1), 87–94. https://doi.org/10.1007/BF00260522
- Gamberg, M. (2000). Contaminants in Yukon country foods.
- Gill, H. K., Lantz, T. C., O'Neill, B., & Kokelj, S. V. (2014). Cumulative Impacts and Feedbacks of a Gravel Road on Shrub Tundra Ecosystems in the Peel Plateau, Northwest Territories, Canada. Arctic Antarctic and Alpine Research, 46(4), 947–961. https://doi.org/10.1657/1938-4246-46. 4.947
- Goyal, R., & Seaward, M. R. D. (1981). Metal Uptake in Terricolous Lichens. New Phytologist, 89(4), 631–645. https://doi.org/10.1111/j.1469-8137.1981.tb02342.x
- Griffiths, R. P., Madritch, M. D., & Swanson, A. K. (2009). The effects of topography on forest soil characteristics in the Oregon Cascade Mountains (USA): Implications for the effects of climate change on soil properties. *Forest Ecology and Management*, 257(1), 1–7. https://doi. org/10.1016/j.foreco.2008.08.010
- Gunther, A. J. (1989). Nitrogen Fixation by Lichens in a Subarctic Alaskan Watershed. *The Bryologist*, 92(2), 202–208. https://doi.org/10.2307/3243946
- Harris, S. A. (1987). Altitude Trends in Permafrost Active Layer Thickness. *Kluane Lake, Yukon Territory. Arctic, 40*(3), 179–183. JSTOR. https://doi.org/10.14430/arctic1764
- Harris, S. A. (1990). Dynamics and Origin of Saline Soils on the Slims River Delta. *Kluane National Park, Yukon Territory. Arctic, 43*(2), 159–175. JSTOR. https://doi.org/10.14430/arctic1607
- He, Z. L., Zhang, M., Yang, X. E., & Stoffella, P. J. (2006). Release Behavior of Copper and Zinc from Sandy Soils. Soil Science Society of America Journal, 70(5), 1699–1707. https://doi.org/10.2136/sssaj2005.0255
- Hoefs, M. (1976). Ecological investigation of dall sheep (Ovis dalli dalli, nelson) and their habitat on sheep mountain, Kluane national Park, Yukon territory. Canada [University of British Columbia]. https://doi.org/10.14288/1.0093879
- Hoefs, M., & Thomson, J. W. (1972). Lichens from the Kluane Game Sanctuary, S.W. yukon Territory. Canadian Field-Naturalist, 86(3), 249–252. https://doi.org/10.5962/p.343602
- Hope, A. S., Fleming, J. B., Stow, D. A., & Aguado, E. (1991). Tussock Tundra Albedos on the North Slope of Alaska: Effects of Illumination, Vegetation Composition, and Dust Deposition. *Journal of Applied Meteorology*, 30(8), 1200–1206. https://doi.org/10.1175/1520-0450(1991) 030<1200:TTAOTN>2.0.CO;2



- Huang, S., Taddei, P., Lawrence, J., Martins, M. A. G., Li, J., & Koutrakis, P. (2021). Trace element mass fractions in road dust as a function of distance from road. *Journal of the Air and Waste Management Association*, 71(2), 137–146. https://doi.org/10.1080/10962247.2020.1834011
 Huck, R., Bryant, R. G., & King, J. (2023). The (mis)identification of high-latitude dust events using remote sensing methods in the Yukon,
- Canada: A sub-daily variability analysis. Atmospheric Chemistry and Physics, 23(11), 6299–6318. https://doi.org/10.5194/acp-23-6299-2023 Hulbert, L. J. (1997). Geology and metallogeny of the Kluane mafic-ultramafic belt, Yukon Territory, Canada: Eastern Wrangellia, a new Ni-Cu-

- Ibanga, I. J., Umoh, N. B., & Iren, O. B. (2008). Effects of Cement Dust on Soil Chemical Properties in the Calabar Environment, Southeastern Nigeria. Communications in Soil Science and Plant Analysis, 39(3–4), 551–558. https://doi.org/10.1080/00103620701826829
- Israel, S., & Cobbett, R. (2008). Kluane Ranges bedrock geology, White River area (Parts of NTS 115F/9, 15 and 16; 115G/12 and 115K/1, 2). (Annual Report Paper YEG2007_12; Yukon exploration and geology 2007, (Eds.), D. S. Emond, L. R. Blackburn, R. P. Hill, & L. H. Weston, Yukon Geological Survey, 153–167. Retrieved from https://ygsftp.gov.yk.ca/publications/yeg/yeg07/12_israel.pdf
- Johnston, F. M., & Johnston, S. W. (2004). Impacts of Road Disturbance on Soil Properties and on Exotic Plant Occurrence in Subalpine Areas of the Australian Alps. Arctic Antarctic and Alpine Research, 36(2), 201–207. JSTOR. https://doi.org/10.1657/1523-0430(2004)036[0201:iordos] 2.0.co;2
- Johnston, S. W. (2001). The influence of aeolian dust deposits on alpine soils in south-eastern Australia. Soil Research, 39(1), 81–88. https://doi.org/10.1071/sr99121

Kabata-Pendias, A. (2010). Trace Elements in Soils and Plants. CRC Press. https://doi.org/10.1201/b10158

- Khan, F., Hayat, Z., Ramzan, M., ShahDr, P. Z., & Hanif, M. (2013). Effect of slope position on physico-chemical properties of eroded soil. Soil and Environment, 31, 22–28.
- Klapstein, S. J., Walker, A. K., Saunders, C. H., Cameron, R. P., Murimboh, J. D., & O'Driscoll, N. J. (2020). Spatial distribution of mercury and other potentially toxic elements using epiphytic lichens in Nova Scotia. *Chemosphere*, 241, 125064. https://doi.org/10.1016/j.chemosphere. 2019.125064
- Krebs, C. J., Boutin, S. A., Boonstra, R., & Kenney, A. J. (2001). Ecosystem dynamics of the boreal forest: The Kluane project. Oxford University Press.
- Lachance, D., Fortin, G., & Dufour Tremblay, G. (2021). Identification et délimitation des milieux humides du Québec méridional—Version décembre 2021. https://www.environnement.gouv.qc.ca/eau/rives/guide-identif-dellimit-milieux-humides.pdf
- Laxton, N. F., Burn, C. R., & Smith, C. A. S. (1996). Productivity of Loessal Grasslands in the Kluane Lake Region, Yukon Territory, and the Beringian "Production Paradox.". Arctic, 49(2), 129–140. JSTOR. https://doi.org/10.14430/arctic1191
- LaZerte, S., & Albers, S. (2018). weathercan: Download and format weather data from Environment and Climate Change Canada. *Journal of Open Source Software*, 3(22), 571. https://doi.org/10.21105/joss.00571
- Li, X., Talbot, J., King, J., & Wang, M. (2023). Effects of road dust on vegetation composition and surface chemistry of three ombrotrophic peatlands in eastern Canada. *Geoderma*, 439, 116665. https://doi.org/10.1016/j.geoderma.2023.116665
- Mahowald, N. M., Muhs, D. R., Levis, S., Rasch, P. J., Yoshioka, M., Zender, C. S., & Luo, C. (2006). Change in atmospheric mineral aerosols in response to climate: Last glacial period, preindustrial, modern, and doubled carbon dioxide climates. *Journal of Geophysical Research*, 111(D10). https://doi.org/10.1029/2005JD006653
- Mandre, M., & Ots, K. (1999). Growth and Biomass Partitioning of 6-Year-Old Spruces under Alkaline Dust Impact. Water, Air, and Soil Pollution, 114(1), 13–25. https://doi.org/10.1023/A:1005048921852
- Marsh, J., Nouvet, S., Sanborn, P., & Coxson, D. (2006). Composition and function of biological soil crust communities along topographic gradients in grasslands of central interior British Columbia (Chilcotin) and southwestern Yukon (Kluane). *Canadian Journal of Botany*, 84(5), 717–736. https://doi.org/10.1139/b06-026
- Martínez, I., Burgaz, A. R., Vitikainen, O., & Escudero, A. (2003). Distribution patterns in the genus Peltigera Willd. The Lichenologist, 35(4), 301–323. https://doi.org/10.1016/S0024-2829(03)00041-0
- Masuda, H. (2018). Arsenic cycling in the Earth's crust and hydrosphere: Interaction between naturally occurring arsenic and human activities. Progress in Earth and Planetary Science, 5(1), 68. https://doi.org/10.1186/s40645-018-0224-3
- McTainsh, G., & Strong, C. (2007). The role of aeolian dust in ecosystems. Geomorphology, 89(1), 39–54. https://doi.org/10.1016/j.geomorph. 2006.07.028
- Meinander, O., Dagsson-Waldhauserova, P., Amosov, P., Aseyeva, E., Atkins, C., Baklanov, A., et al. (2022). Newly identified climatically and environmentally significant high-latitude dust sources. *Atmospheric Chemistry and Physics*, 22(17), 11889–11930. https://doi.org/10.5194/ acp-22-11889-2022
- Miadlikowska, J., & Lutzoni, F. (2000). Phylogenetic Revision of the Genus Peltigera (Lichen-Forming Ascomycota) Based on Morphological, Chemical, and Large Subunit Nuclear Ribosomal DNA Data. *International Journal of Plant Sciences*, 161(6), 925–958. https://doi.org/10. 1086/317568
- Moorhead, D. L., Linkins, A. E., & Everett, K. R. (1996). Road Dust Alters Extracellular Enzyme Activities in Tussock Tundra Soils, Alaska, U.S. A. Arctic and Alpine Research, 28(3), 346–351. JSTOR. https://doi.org/10.2307/1552114

Muhs, D. R. (2013). The geologic records of dust in the Quaternary. *Aeolian Research*, 9, 3–48. https://doi.org/10.1016/j.aeolia.2012.08.001 Muhs, D. R., Budahn, J. R., Prospero, J. M., & Carey, S. N. (2007). Geochemical evidence for African dust inputs to soils of western Atlantic

islands: Barbados, the Bahamas, and Florida. Journal of Geophysical Research, 112(F2). https://doi.org/10.1029/2005JF000445

Muller, J. E. (1967). Kluane lake map-area, Yukon territory: (115G, 115F E1/2). Geological Survey of Canada.

- Müllerová, J., Vítková, M., & Vítek, O. (2011). The impacts of road and walking trails upon adjacent vegetation: Effects of road building materials on species composition in a nutrient poor environment. *Science of the Total Environment*, 409(19), 3839–3849. https://doi.org/10.1016/j. scitotenv.2011.06.056
- Murphy, J., & Riley, J. P. (1962). A modified single solution method for the determination of phosphate in natural waters. *Analytica Chimica Acta*, 27, 31–36. https://doi.org/10.1016/S0003-2670(00)88444-5
- Myers-Smith, I. H., Arnesen, B. K., Thompson, R. M., & III, F. S. C. (2006). Cumulative impacts on Alaskan arctic tundra of a quarter century of road dust. *Écoscience*, 13(4), 503–510. https://doi.org/10.2980/1195-6860(2006)13[503;CIOAAT]2.0.CO;2
- Naeth, M. A., & Wilkinson, S. R. (2008). Lichens as biomonitors of air quality around a diamond mine, northwest territories, Canada. Journal of Environmental Quality, 37(5), 1675–1684. https://doi.org/10.2134/jeq2007.0090
- Nagy, M. L., Johansen, J. R., Clair, L. L. S., & Webb, B. L. (2005). Recovery patterns of microbiotic soil crusts 70 years after arsenic contamination. *Journal of Arid Environments*, 63(1), 304–323. https://doi.org/10.1016/j.jaridenv.2005.03.029
- Nash, T. H. (2008). Nutrients, elemental accumulation, and mineral cycling. III. In T. H. Nash (Ed.), *Lichen biology* (2nd ed., pp. 234–251). Cambridge University Press. https://doi.org/10.1017/CBO9780511790478.013

PGE metallogenic terrane. Geological Survey of Canada.



- Neff, J. C., Ballantyne, A. P., Farmer, G. L., Mahowald, N. M., Conroy, J. L., Landry, C. C., et al. (2008). Increasing eolian dust deposition in the western United States linked to human activity. *Nature Geoscience*, 1(3), 189–195. https://doi.org/10.1038/ngeo133
- Neitlich, P. N., Berryman, S., Geiser, L. H., Mines, A., & Shiel, A. E. (2022). Impacts on tundra vegetation from heavy metal-enriched fugitive dust on National Park Service lands along the Red Dog Mine haul road, Alaska. *PLoS One*, 17(6), e0269801. https://doi.org/10.1371/journal. pone.0269801
- Neitlich, P. N., Hoef, J. M. V., Berryman, S. D., Mines, A., Geiser, L. H., Hasselbach, L. M., & Shiel, A. E. (2017). Trends in spatial patterns of heavy metal deposition on national park service lands along the Red Dog Mine haul road, Alaska, 2001–2006. PLoS One, 12(5), e0177936. https://doi.org/10.1371/journal.pone.0177936
- Nickling, W. G. (1978). Eolian sediment transport during dust storms: Slims River Valley, Yukon Territory. *Canadian Journal of Earth Sciences*, 15(7), 1069–1084. https://doi.org/10.1139/e78-114
- Normand, S., Treier, U. A., Randin, C., Vittoz, P., Guisan, A., & Svenning, J.-C. (2009). Importance of abiotic stress as a range-limit determinant for European plants: Insights from species responses to climatic gradients. *Global Ecology and Biogeography*, 18(4), 437–449. https://doi.org/ 10.1111/j.1466-8238.2009.00451.x
- Okin, G. S., Gillette, D. A., & Herrick, J. E. (2006). Multi-scale controls on and consequences of aeolian processes in landscape change in arid and semi-arid environments. *Journal of Arid Environments*, 65(2), 253–275. https://doi.org/10.1016/j.jaridenv.2005.06.029
- Okin, G. S., Mahowald, N., Chadwick, O. A., & Artaxo, P. (2004). Impact of desert dust on the biogeochemistry of phosphorus in terrestrial ecosystems. *Global Biogeochemical Cycles*, 18(2). https://doi.org/10.1029/2003GB002145
- Orton, T. G., Rawlins, B. G., & Lark, R. M. (2009). Using measurements close to a detection limit in a geostatistical case study to predict selenium concentration in topsoil. *Geoderma*, 152(3), 269–282. https://doi.org/10.1016/j.geoderma.2009.06.014
- Palmqvist, K., Dahlman, L., Valladares, F., Tehler, A., Sancho, L. G., & Mattsson, J.-E. (2002). CO2 exchange and thallus nitrogen across 75 contrasting lichen associations from different climate zones. *Oecologia*, 133(3), 295–306. https://doi.org/10.1007/s00442-002-1019-0
- Parviainen, A., Casares-Porcel, M., Marchesi, C., & Garrido, C. J. (2019). Lichens as a spatial record of metal air pollution in the industrialized city of Huelva (SW Spain). Environmental Pollution, 253, 918–929. https://doi.org/10.1016/j.envpol.2019.07.086
- Paulsrud, P., Rikkinen, J., & Lindblad, P. (2001). Field investigations on cyanobacterial specificity in Peltigera aphthosa. *New Phytologist*, *152*(1), 117–123. https://doi.org/10.1046/j.0028-646x.2001.00234.x
- Pisani, T., Munzi, S., Paoli, L., Bačkor, M., & Loppi, S. (2011). Physiological effects of arsenic in the lichen Xanthoria parietina (L.) Th. Fr. Chemosphere, 82(7), 963–969. https://doi.org/10.1016/j.chemosphere.2010.10.079
- Posit team. (2023). RStudio: Integrated development environment for R. [Computer software]. Posit Software, PBC. Retrieved from http://www.posit.co/
- Pouillé, S. (2024). Concentrations of trace elements measured in lichens and soils at Lhù'ààn Mân' (Kluane Lake, Yukon territory, Canada) in August 2021. [dataset]. PANGAEA. Retrieved from https://doi.pangaea.de/10.1594/PANGAEA.965325
- QGIS Development Team. (2023). QGIS Geographic Information System. [Computer software]. Retrieved from http://qgis.osgeo.org. Open Source Geospatial Foundation.
- Ratnaike, R. N. (2003). Acute and chronic arsenic toxicity. Postgraduate Medical Journal, 79(933), 391–396. https://doi.org/10.1136/pmj.79. 933.391
- Ravi, S., D'Odorico, P., Breshears, D. D., Field, J. P., Goudie, A. S., Huxman, T. E., et al. (2011). Aeolian Processes and the Biosphere. *Reviews of Geophysics*, 49(3). https://doi.org/10.1029/2010RG000328
- R Core Team. (2022). R: A language and environment for statistical computing. [Computer software]. R Foundation for Statistical Computing. Retrieved from https://www.R-project.org/
- Riget, F., Asmund, G., & Aastrup, P. (2000). The use of lichen (Cetraria nivalis) and moss (Rhacomitrium lanuginosum) as monitors for atmospheric deposition in Greenland. Science of the Total Environment, 245(1), 137–148. https://doi.org/10.1016/S0048-9697(99)00439-8
- Riley, J. L. (1989). Laboratory methods for testing peat: Ontario peatland inventory project. Retrieved from https://www.osti.gov/etdeweb/biblio/ 6912751
- Santelmann, M. V., & Gorham, E. (1988). The Influence of Airborne Road Dust on the Chemistry of Sphagnum Mosses. Journal of Ecology, 76(4), 1219–1231. JSTOR. https://doi.org/10.2307/2260644
- Sauer, D., Schellmann, G., & Stahr, K. (2007). A soil chronosequence in the semi-arid environment of Patagonia (Argentina). Catena, 71(3), 382–393. https://doi.org/10.1016/j.catena.2007.03.010
- Seaward, M. R. D. (2008). Environmental role of lichens. III. In T. H. Nash (Ed.), *Lichen biology* (2nd ed., pp. 274–298). Cambridge University Press. https://doi.org/10.1017/CBO9780511790478.015
- Shugar, D. H., Clague, J. J., Best, J. L., Schoof, C., Willis, M. J., Copland, L., & Roe, G. H. (2017). River piracy and drainage basin reorganization led by climate-driven glacier retreat. *Nature Geoscience*, 10(5), 370–375. Article 5. https://doi.org/10.1038/ngeo2932
- Simonetti, A., Gariépy, C., & Carignan, J. (2003). Tracing sources of atmospheric pollution in Western Canada using the Pb isotopic composition and heavy metal abundances of epiphytic lichens. *Atmospheric Environment*, 37(20), 2853–2865. https://doi.org/10.1016/S1352-2310(03) 00210-3
- Singh, S. M., Sharma, J., Gawas-Sakhalkar, P., Upadhyay, A. K., Naik, S., Pedneker, S. M., & Ravindra, R. (2013). Atmospheric deposition studies of heavy metals in Arctic by comparative analysis of lichens and cryoconite. *Environmental Monitoring and Assessment*, 185(2), 1367–1376. https://doi.org/10.1007/s10661-012-2638-5
- Sloof, J. E. (1995). Lichens as quantitative biomonitors for atmospheric trace-element deposition, using transplants. *Atmospheric Environment*, 29(1), 11–20. https://doi.org/10.1016/1352-2310(94)00221-6
- Steinnes, E., & Friedland, A. J. (2006). Metal contamination of natural surface soils from long-range atmospheric transport: Existing and missing knowledge. *Environmental Reviews*, 14(3), 169–186. https://doi.org/10.1139/a06-002
- Swap, R., Garstang, M., Greco, S., Talbot, R., & Kållberg, P. (1992). Saharan dust in the Amazon Basin. Tellus B: Chemical and Physical Meteorology, 44(2), 133–149. https://doi.org/10.3402/tellusb.v44i2.15434
- Tanaka, T. (1988). Distribution of arsenic in the natural environment with emphasis on rocks and soils. Applied Organometallic Chemistry, 2(4), 283–295. https://doi.org/10.1002/aoc.590020403
- Testolin, R., Attorre, F., & Jiménez-Alfaro, B. (2020). Global distribution and bioclimatic characterization of alpine biomes. *Ecography*, 43(6), 779–788. https://doi.org/10.1111/ecog.05012
- Tobias, M., & Niinemets, Ü. (2010). Acclimation of photosynthetic characteristics of the moss Pleurozium schreberi to among-habitat and withincanopy light gradients. *Plant Biology*, *12*(5), 743–754. https://doi.org/10.1111/j.1438-8677.2009.00285.x
- Tyler, G. (1989). Uptake, retention and toxicity of heavy metals in Lichens. Water, Air, and Soil Pollution, 47(3), 321-333. https://doi.org/10. 1007/BF00279330



- Vetter, M. A. (2000). Grasslands of the Aishihik-Sekulmun Lakes Area. Yukon Territory, Canada. Arctic, 53(2), 165–173. JSTOR. https://doi.org/ 10.14430/arctic847
- Wackernagel, H. (2003). Multivariate geostatistics: An introduction with applications. Springer Science and Business Media.
- Walker, D. A. (1996). Disturbance and Recovery of Arctic Alaskan Vegetation. In J. F. Reynolds & J. D. Tenhunen (Eds.), Landscape function and disturbance in arctic tundra (pp. 35–71). Springer. https://doi.org/10.1007/978-3-662-01145-4_3
- Walker, D. A., & Everett, K. R. (1987). Road Dust and Its Environmental Impact on Alaskan Taiga and Tundra. Arctic and Alpine Research, 19(4), 479–489. JSTOR. https://doi.org/10.2307/1551414
- Whittinghill, K. A., & Hobbie, S. E. (2012). Effects of pH and calcium on soil organic matter dynamics in Alaskan tundra. *Biogeochemistry*, 111(1), 569–581. https://doi.org/10.1007/s10533-011-9688-6
- Widory, D., Vautour, G., & Poirier, A. (2018). Atmospheric dispersion of trace metals between two smelters: An approach coupling lead, strontium and osmium isotopes from bioindicators. *Ecological Indicators*, *84*, 497–506. https://doi.org/10.1016/j.ecolind.2017.09.003
- Wojtuń, B., Samecka-Cymerman, A., Kolon, K., Kempers, A. J., & Skrzypek, G. (2013). Metals in some dominant vascular plants, mosses, lichens, algae, and the biological soil crust in various types of terrestrial tundra, SW Spitsbergen, Norway. *Polar Biology*, 36(12), 1799–1809. https://doi.org/10.1007/s00300-013-1399-0
- Zdanowicz, C., Zheng, J., Klimenko, E., & Outridge, P. M. (2017). Mercury and other trace metals in the seasonal snowpack across the subarctic taiga-tundra ecotone, Northwest Territories, Canada. Applied Geochemistry, 82, 63–78. https://doi.org/10.1016/j.apgeochem.2017.04.011